# Why test study protocol: a UK-wide audit using the Primary Care Academic Collaborative to explore the reasons for primary care testing

Alexander Burrell<sup>1</sup>, Polly Duncan<sup>1</sup>, Ian Bennett-Britton<sup>2</sup>, Sam Hodgson<sup>2</sup>, Samuel W D Merriel<sup>3</sup>, Salman Waqar<sup>5</sup>, Penny F Whiting<sup>1</sup>, Jessica Watson<sup>1</sup>, on behalf of the Primary Care Academic Collaborative (PACT)

- 1. Centre for Academic Primary Care, University of Bristol
- 2. Primary Care Research Centre, School of Primary Care, Population Sciences and Medical Education, University of Southampton
- 3. Exeter Collaboration for Academic Primary Care (APEx), Exeter Medical School, University of Exeter
- 4. Nuffield Department of Primary Care Health Sciences, University of Oxford
- 5. Bristol Population Health Science Institute, University of Bristol

# **Corresponding author:**

Dr Jessica Watson, Clinical Lecturer, Centre for Academic Primary Care, Population Health Sciences, University of Bristol
Jessica.Watson@bristol.ac.uk

Word count: 2346

# Abstract

#### **Background**

The number of blood tests done in primary care has been increasing over the last 20 years. Some estimates suggest that up to a quarter of these tests may not have been needed. This could lead to a cascade effect of further investigations, appointments, or referrals, as well as anxiety for patients, increased workload and costs to the health service. To better understand the impact and sequelae of blood tests on patients, we need to know why blood tests are requested and what is done with the results.

#### **Aims**

To explore who orders blood tests and why, and how test results are actioned in primary care.

#### **Design & Setting**

Retrospective audit of electronic health records in general practices across the UK.

#### Method

The Primary care Academic CollaboraTive (PACT), a UK-wide network of primary care health professionals, will be utilised to collect data from individual practices. PACT members will be asked to review the electronic health records of 50 patients who had recent blood tests in their practice, and manually extract anonymised data on who requested the test, the indication, the result, and subsequent actions. Data will also be collected from PACT members to assess the feasibility of the collaborative model.

# Conclusion

PACT offers a unique opportunity to extract clinical data which cannot otherwise be obtained. Understanding the indications for tests will help identify priority areas for research to optimise testing and patient safety in primary care.

#### Keywords

Primary Health Care; General Practice; Clinical Laboratory Techniques; Clinical Decision-Making; Collaborative Research

#### How this fits in

The number of blood tests done in GP surgeries has been increasing without a clear corresponding increase in diagnosis of disease. Estimates suggest up to a quarter of tests may not be necessary. This study, using a novel collaborative research model, will collect data from practices across the United Kingdom on why blood tests are done and how the results are used. This will help identify priority areas for future research to optimise testing in the community.

#### Introduction

Routine data from primary care electronic health records has shown large increases in the use of blood tests in UK primary care over the past two decades, <sup>1,2</sup> and significant variation in testing rates between GP practices. This rise in testing has taken place in the context of significant uncertainty and lack of evidence to determine which tests are 'necessary', with guidelines for chronic disease monitoring mostly based on expert opinion, <sup>4</sup> and many reasons for testing in primary care falling outside of clinical guidelines. Estimates have suggested that 25% of primary care pathology testing might be 'unnecessary', <sup>5</sup> with research demonstrating unwarranted variation and overuse of specific tests including thyroid function tests, liver function tests, prostate specific antigen tests and vitamin D. This may lead to further blood tests, imaging, GP appointments and referrals, a process sometimes referred to as the "cascade effect", <sup>7</sup> with implications for GP workload, patient anxiety and healthcare costs. The concept of the cascade effect has been around for over 30 years, <sup>8</sup> but is rarely measured, <sup>9</sup> and the overall frequency and implications of cascade testing on primary care workload is unknown.

Reduction in unwarranted variation in testing rates has been frequently cited as an aim,<sup>10</sup> particularly in the current context of rising workload,<sup>11</sup> a primary care workforce crisis, and concerns about socioeconomic inequalities in health. A prerequisite to achieving this aim is to first understand the rationale for blood testing in primary care, and the outcomes of testing. This information cannot be obtained easily from current Electronic Health Record data.

Studies quantifying failures in test result follow up have been systematically reviewed, with between 6.8% and 62% of laboratory tests reportedly not followed-up in US settings, and no relevant UK research identified. Surveys and qualitative research have demonstrated that most UK general practices rely on patients contacting the practice for their test results, with a lack of fail-safe mechanisms. Errors associated with failures in filing, communicating and actioning of abnormal results can lead to delayed and missed diagnoses, which are a common reason for litigation in primary care. The World Health Organisation has identified that rates of test result follow up are

'suboptimal', leading to serious lapses in care.<sup>18</sup> Tools have been developed for individual practices to audit and improve blood test handling,<sup>19</sup> yet dissemination of this shared learning between GP practices is challenging.

The Primary care Academic CollaboraTive (PACT) is a new UK-wide network of primary care health professionals from England, Wales, Scotland and Northern Ireland, who collectively take part in primary care research and quality improvement (QI) projects that seek to improve patient care. There are over 650 current members of PACT, who have been recruited via social media, electronic newsletters and promotion at primary care conferences and meetings. PACT membership is free and voluntary, members receive a monthly newsletter, with no expectation to participate in PACT activities. Current membership comprises 55% GP trainees, 21% GPs, 7% allied health professionals; the remainder are students and non-clinical researchers (who are not eligible to participate in the Why Test study). PACT members will collect data for projects in their individual practices and the data will be combined to increase the power and generalisability of the results. Practice-level data, benchmarked against other practices taking part, can be used by PACT members to identify areas for QI.

PACT offers a unique opportunity to extract clinical data which cannot otherwise be obtained, including reasons for blood testing, actioning of abnormal results and the cascade effects of blood tests in primary care. Understanding the reasons why tests are currently performed will help identify priority areas for future research to optimise testing in primary care, by identifying clinical areas associated with high volumes of tests, and low yield of abnormal results.

#### Aims

The primary aim of the study is to explore who orders blood tests and why, and how test results are actioned in primary care. As this is one of the first projects using the PACT network the secondary aim is to assess the feasibility of using a new network of trainees and health professionals in primary care to carry out high quality research.

The overall aims of this study are broken down into the following specific objectives

- 1. Why are laboratory tests requested in primary care?
  - a. What proportion are for screening, monitoring and diagnosis?
  - b. For diagnostic testing, what are the symptoms which lead to blood testing?
- 2. Who is responsible for ordering laboratory tests in primary care?
  - a. What proportion are requested by GPs, nurses and other allied health professionals?

- 3. How are laboratory test results actioned in primary care?
  - a. What proportion lead to follow on blood tests and GP appointments?
  - b. What proportion of abnormal test results are actioned as intended by the clinician filing the results?
- 4. What proportion of tests yield abnormal results?
  - a. Does this vary depending on the symptoms/reasons for testing?
- 5. What proportion of tests lead to a change in management?

#### Method

#### Recruitment and sampling

The study will recruit PACT members from across the UK including foundation doctors, GP trainees, GPs, nurses and allied health professionals working in primary care (hereafter 'PACT members'). All current PACT members will be invited to participate in the study. We will ask PACT members to complete an online expression of interest and consent form. A GP partner or practice manager from the PACT member's GP practice will then be required to complete a practice agreement form. A summary of recruitment and consent to the study is shown in **Figure 1**.

We will use purposive sampling to recruit the first 5-10 pilot GP practices, aiming to include a range of PACT team members and a range of electronic health records systems (EMIS, SystmOne and Vision). These pilot practices will be used to identify any problems with the data collection tools prior to the wider rollout. To evaluate the pilot, participating PACT members will be invited to weekly online drop-in meetings with the research team to discuss any issues and ask questions. PACT members will also complete a short online survey on completion of the pilot.

Once data from the first 10 practices has been successfully extracted, we invite other practices that have expressed an interest to take part. We aim to recruit at least 50 practices, with 50 patients per practice providing a total of 2500 patients in the study. If more practices are recruited then we will have greater precision in our estimates.

Service support costs will be available to cover GP supervision and administrative support to the PACT member, in line with AcoRD guidelines.<sup>20</sup>

#### **Training**

PACT members will be required to watch two short training videos, and code three fictitious clinical cases using REDCap prior to commencing data collection (see **Supplementary Box 1**). Training

materials for participants will be amended following feedback from the pilot practices, to ensure they reflect common areas of uncertainty encountered during the piloting. A pass mark of >70% for each of the three test cases will be required; PACT members scoring =<70% will need to repeat the training. Inter-rater reliability will be assessed using the three test cases; we will compare average training scores between GPs, trainees and allied health professionals to measure the variability of coding amongst different these professional groups.

#### Data collection

Each PACT member will be asked to collect anonymised data by reviewing the GP electronic health records of 50 patients. Any patient aged >=18 years old, having a blood test in primary care during April 2021 will be eligible for inclusion. This period was chosen pragmatically to capture usual practice following the early waves of the covid-19 pandemic, and to allow sufficient time for follow up prior to the blood bottle shortages in August 2021. We will ask PACT members to identify any unusual circumstances within the practice that may have affected testing within the timeframe sampled. Pregnant women and children will be excluded due to biochemical differences in reference ranges for routine bloods. Each PACT member will use a pre-defined search strategy to identify 50 randomly selected eligible patients.

The PACT member will then review the notes of each patient to manually extract anonymised data into a REDCap database. Data collected will include patient demographics, reasons for testing and the clinician's actioning of test results (see **Supplementary Table 1**). If the reason for testing is coded as 'symptoms/diagnosis', this will be subcategorised using the International Classification for Primary Care (ICPC2). The variables for data collection have been developed iteratively using evidence from previous literature reviews,<sup>21</sup> questionnaire studies<sup>22,23</sup> and pre-piloting by members of the research team. Based on pre-pilot work we estimate it will take 5-10 minutes per patient. The variables for data collection (**Supplementary Table 1**) will be modified following piloting as required.

# Analysis plans

Using a flow diagram, we will examine the number of PACT members who expressed an interest in taking part, the number that were invited and the number that completed the project.

Descriptive analyses will be used to describe:

1) Characteristics of participating PACT members (age, gender, ethnicity, healthcare professional type) and their GP practices (list size, index of multiple deprivation, region)

# 2) Proportions of tests:

- a. requested for screening, monitoring and diagnostic purposes
- b. requested by GPs, nurses and allied health professionals
- c. leading to change in management (new medication, diagnosis, follow on tests, referrals)
- d. actioned as intended (and conversely the proportion of tests which were not actioned and lead to harm or potential harm to patients)
- e. where there is documented evidence that patients were informed of their results

We will use logistic regression analyses to measure associations between reasons for testing (exposure variable) and frequency of abnormal results (outcome), adjusted for age and gender. The primary outcome will be tests coded as 'abnormal' by PACT members, with a sensitivity analysis to include both 'abnormal' and 'borderline' results. Secondary outcomes will be 'actionable results', defined as tests which trigger an action from the GP coding the test results (eg repeat testing, follow up appointment or telephone call), and 'practice changing results', defined as tests which lead to a change in clinical management (eg new medication, new diagnosis, referral etc). This will give an overview of which indications for testing generate a higher yield of abnormal results.

#### PACT member involvement and engagement

To evaluate the feasibility of the PACT network for carrying out high quality research projects we will collect data on the demographic characteristics of participating PACT members and characteristics of participating GP practices. We will also collect and collate feedback from participating PACT members and information about QI activities and learning events triggered by the project. PACT members will be able to contribute to the research process via the secure NIHR-Learn platform, and by optional attendance at open meetings. PACT members who take part in the study will be coauthors or collaborators on the final publication (depending on journal guidelines). We will offer the opportunity for participating PACT members to propose and undertake additional analyses using the data to answer questions of interest to frontline clinicians, in order to ensure that this is a truly collaborative project. This could include for example sub-analyses looking at specific tests of interest to PACT clinicians such as liver function tests.

# Patient and Public Involvement and Engagement (PPIE)

A PPIE group was consulted for the purpose of this study, comprising five patients with personal experience of undergoing tests in primary care. They shared experiences which resonated with the

research objectives, and were particularly interested in not only the reasons for testing but communication of test results. This led to changes in the research objectives and data collection tools to capture test communication. The PPIE group also provided input into the lay summary and were involved in discussions regarding the ethical issues of the proposed data collection methods.

#### Discussion

This is a UK-wide study utilising a novel collaborative primary care research network to better understand who orders tests, why, and how the results are used in primary care.

#### **Strengths and Limitations**

This is one of the first studies to test the feasibility of using a collaborative model to conduct research within a primary care setting. Collaborative networks provide opportunities for non-academic clinicians to be involved in research and have been utilised in other specialties to conduct high impact studies. <sup>24</sup> The PACT model will allow extraction of data which cannot be measured using routine data from electronic health record databases, such as clinical rationale for testing. Using clinicians to extract data allows subjective clinical opinions to be explored, for example 'in your clinical opinion, were the tests necessary?' - we will use the results of these subjective questions to explore how clinicians' views vary, rather than viewing them as providing definitive data. Practices which choose to participate might be more likely to have an interest in QI or more resources for participation, this should be taken into account when interpreting the results. Similar methods of data collection have been used successfully by the National Cancer Diagnosis Audit. <sup>25</sup> We will measure inter-rater reliability to ensure data-collection methods are rigorous and reproducible. If feasible, the PACT model can be used to address further primary care research questions of importance to practicing clinicians.

# **Implications for Practice**

On a local level, a practice report containing a summary of the descriptive statistics will be emailed to participating PACT members, including anonymised practice-level data and combined practice data to benchmark against. PACT members will be able to use the data from their own practice to identify areas for QI around filing, communicating and actioning blood test results. Ideas for QI will be collated and shared with participating practices. This is important as analysis of UK medical protection organisation's database have demonstrated that hazards in primary care systems of coding and communicating laboratory tests are a common cause of malpractice and litigation. <sup>26,27</sup>

On a national level, this project will be important to help identify which clinical areas generate the largest volumes of blood testing, and highlight which symptoms are associated with the lowest yield of abnormal results. This will be important for developing future research and QI activities to optimise laboratory testing in primary care.

# Funding

The study has been adopted by NIHR Applied Research Collaborative West and funded by Research Capability Funding from the Bristol, North Somerset and South Gloucestershire Clinical Commissioning Group (RCF20/21-1-1JW & RCF21/22-1JW).

# Ethical approval

This study has received favourable ethical opinion from University of Bristol Faculty of Health Sciences Research Ethics Committee (ref 116947).

# Acknowledgements

Thank you to the participants in the Patient and Public Involvement and Engagement group whose input led to numerous direct changes to the protocol and analysis plan, and to Chris Salisbury for advice and guidance.

# Competing interests

SH is a member of the BJGP Open Editorial Board; he had no involvement in the peer review process or decision on this manuscript. SWDM is Associate Editor of the BJGP; he had no involvement in the peer review process or decision on this manuscript.

Figure 1: Recruitment and consent flowchart

#### Advertise the study

Via PACT network, social media, GP teaching newsletters, RCGP and Next Generation **GP Networks** 

#### **Expression of interest**

Link on PACT website with information about study. Click through to online expression of interest form on REDCap, including short questionnaire to collect demographic details of GP practice and an electronic consent form

# Practice agreement

On line practice agreement form on REDCap, for completion by GP partner or practice

Practice selection to take part in study
Practices with a range of computer systems (EMIS, SystemOne, Vision) and a range of
PACT members (trainees, GPs, allied health professionals) selected to participate in

Recruitment to main study
All practices which expressed an interest will be invited to participate

# References

- 1. O'Sullivan JW, Stevens S, Hobbs FDR, et al. Temporal trends in use of tests in UK primary care, 2000-15: retrospective analysis of 250 million tests. *BMJ*. 2018;363.
- 2. Busby J, Schroeder K, Woltersdorf W, et al. Temporal growth and geographic variation in the use of laboratory tests by NHS general practices: using routine data to identify research priorities. *Br J Gen Pract.* 2013;63(609):e256-266.
- 3. O'Sullivan JW, Stevens S, Oke J, et al. Practice variation in the use of tests in UK primary care: a retrospective analysis of 16 million tests performed over 3.3 million patient years in 2015/16. BMC Med. 2018;16(1):229.
- 4. Elwenspoek MMC, Patel R, Watson JC, Whiting P. Are guidelines for monitoring chronic disease in primary care evidence based? *BMJ.* 2019;365:l2319.
- 5. Report of the Second Phase of the Review of NHS Pathology Services in England. Chaired by Lord Carter of Coles. 2008.
- 6. M. Zhi, E.L. Ding, J. Theisen-Toupal, J. Whelan, R. Arnaout, The landscape of inappropriate laboratory testing: a 15-year meta-analysis, *PloS one*. 8(11) (2013) e78962.
- 7. Deyo RA. Cascade effects of medical technology. *Annu Rev Public Health.* 2002;23:23-44.
- 8. Mold JW, Stein HF. The cascade effect in the clinical care of patients. *N Engl J Med.* 1986;314(8):512-514.
- 9. Watson J SC, Whiting P, Banks J, Pyne Y, Hamilton W. Added value and cascade effects of inflammatory marker tests in UK primary care: a cohort study from the Clinical Practice Research Datalink. *Br J Gen Pract.* 2019; 69 (684): e470-e478.
- 10. Fryer AA, Smellie WSA. Managing demand for laboratory tests: a laboratory toolkit. *J Clin Pathol.* 2013;66(1):62-72.
- 11. Hobbs FDR, Bankhead C, Mukhtar T, et al. Clinical workload in UK primary care: a retrospective analysis of 100 million consultations in England. *Lancet*. 2016; 387(10035):2323-2330.
- 12. Callen JL, Westbrook JI, Georgiou A, Li J. Failure to follow-up test results for ambulatory patients: a systematic review. *J Gen Intern Med.* 2012;27(10):1334-1348.
- 13. Litchfield I, Bentham L, Lilford R, McManus RJ, Hill A, Greenfield S. Test result communication in primary care: a survey of current practice. *BMJ Qual Saf.* 2015;24(11):691-699
- 14. Litchfield IJ, Bentham LM, Lilford RJ, Greenfield SM. Test result communication in primary care: clinical and office staff perspectives. *Fam Pract*. 2014;31(5):592-597.
- 15. Litchfield I, Bentham L, Hill A, McManus RJ, Lilford R, Greenfield S. Routine failures in the process for blood testing and the communication of results to patients in primary care in the UK: a qualitative exploration of patient and provider perspectives. *BMJ Qual Saf*. 2015;24(11):681-690.
- 16. Kostopoulou O, Delaney BC, Munro CW. Diagnostic difficulty and error in primary care--a systematic review. *Fam Pract.* 2008;25(6):400-413.
- 17. Singh H, Schiff GD, Graber ML, Onakpoya I, Thompson MJ. The global burden of diagnostic errors in primary care. *BMJ Qual Saf.* 2017;26(6):484-494.
- Summary of the evidence on patient safety: Implications for research. World alliance for patient safety: WHO: 2008. Available at: <a href="https://www.who.int/publications/i/item/9789241596541">https://www.who.int/publications/i/item/9789241596541</a> (accessed 22/04/2022).
- 19. Health Improvement Scotland. Scottish Patient Safety Programme Primary Care: Results Handling. Available at: <a href="https://ihub.scot/improvement-programmes/scottish-patient-safety-programme-spsp/spsp-programmes-of-work/spsp-primary-care/deteriorating-patient-in-primary-care/results-handling/">https://ihub.scot/improvement-programmes/scottish-patient-safety-programme-spsp/spsp-programmes-of-work/spsp-primary-care/deteriorating-patient-in-primary-care/results-handling/</a> (accessed 22/04/2022)
- 20. Department of Health. Attributing the costs of health and social care Research & Development (AcoRD) 2012. Available at:

- https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attachment data/file/351182/AcoRD Guidance for publication May 2012.pdf (accessed 22/04/2022)
- 21. Whiting P, Toerien M, de Salis I, et al. A review identifies and classifies reasons for ordering diagnostic tests. *J Clin Epidemiol*. 2007;60(10):981-989.
- 22. Houben PH, van der Weijden T, Winkens RA, Grol RP. Cascade effects of laboratory testing are found to be rare in low disease probability situations: prospective cohort study. *J Clin Epidemiol*. 2010;63(4):452-458.
- 23. Houben PH, Winkens RA, van der Weijden T, Vossen RC, Naus AJ, Grol RP. Reasons for ordering laboratory tests and relationship with frequency of abnormal results. *Scand J Prim Health Care*. 2010;28(1):18-23.
- 24. Jamjoom AA, Phan PN, Hutchinson PJ, Kolias AG. Surgical trainee research collaboratives in the UK: an observational study of research activity and publication productivity. *BMJ Open*. 2016;6(2):e010374.
- 25. Swann R, McPhail S, Witt J, Shand B, Abel GA, Hiom S, Rashbass J, Lyratzopoulos G, Rubin G. Diagnosing cancer in primary care: results from the National Cancer Diagnosis Audit. *Br J Gen Pract*. 2018; 68 (666): e63-e72. DOI: 10.3399/bjgp17X694169
- 26. P. Bowie, J. Price, N. Hepworth, M. Dinwoodie, J. McKay, System hazards in managing laboratory test requests and results in primary care: medical protection database analysis and conceptual model, *BMJ Open.* 2015; 5(11):e008968.
- 27. D. Baylis, J. Price, P. Bowie, Content analysis of 50 clinical negligence claims involving test results management systems in general practice, *BMJ Open Qual.* 2018;7(4):e000463.